\noindent \textbf{Determine whether \(<<pro_value>>\) is a solution to the equation \(<<equation>>\):}
\vspace{4pt}  % Ensure spacing between problem statement and solution

\noindent
\renewcommand{\arraystretch}{1.3} % Adjust line spacing in the aligned environment
\begin{tabular}{@{}p{0.45\linewidth}@{}p{0.45\linewidth}@{}}
    \(\begin{aligned}
        \text{LHS} &= <<LHS>> \\
                   &= <<LHSsub>> \\
                   &= <<LHSval>>
    \end{aligned}\) &
    \(\begin{aligned}
        \text{RHS} &= <<RHS>>\\
                   & \\
                   &
    \end{aligned}\)
\end{tabular}
\renewcommand{\arraystretch}{1.0} % Adjust line spacing in the aligned environment
\vspace{2pt}  % Optional spacing

\noindent \(\therefore\) Since \(\text{LHS} <<side_equality>> \text{RHS}\), \(<<pro_value>>\) <<is_a_sol>> a solution to the equation.